Author's response to reviews

Title: The Riddle of Multinucleated "Floret-Like" Giant Cells and their Detection in an Extensive Gluteal Neurofibroma

Authors:

Katrin Merz (Katrin.Merz@uk-erlangen.de)
Sora De Kerviler (sora.dekerviler@kssg.ch)
Istvan Vajtai (istvan.vajtai@pathology.unibe.ch)
Mihai Constantinescu (Mihai.Constantinescu@insel.ch)

Version: 2 Date: 15 January 2013

Author's response to reviews:

Up to now there were only formal revisions done to the manuscript